Author’s response to reviews

Title: The role of emotion in clinical decision making: an integrative literature review

Authors:

Desirée Kozlowski (desiree.kozlowski@scu.edu.au; dk@pleasurelab.com.au)

Marie Hutchinson (Marie.Hutchinson@scu.edu.au)

John Hurley (John.Hurley@scu.edu.au)

Joanne Rowley (Joanne.Rowley@scu.edu.au)

Joanna Sutherland (jsutherland@bongil.com.au)

Version: 2 Date: 07 Nov 2017

Author’s response to reviews:

Editor’s request for minor revisions:

Reidar Tyssen, M.D., Ph.D.:

1) The Result section of the Abstract should include your identified major themes and sub-themes (arbitrary) on the expense of only description of studies examined and words in lengthy conclusion (could be only a sentence).

We have added themes and sub-themes to the results section of the Abstract and removed a sentence from the conclusion section. Word count is now 301.

2) The conclusion of the discussion should stick only to the main findings (first paragraph). I suggest that you move the last part of this conclusion, second para with new references, to the end of the discussion.

Thank you. We have taken your advice on this.